Author’s response to reviews

Title: Return of Genetic and Genomic Research Findings: Experience of a Pediatric Biorepository

Authors:

Tanya Papaz (tanya.daljevic@sickkids.ca)
Eriskay Liston (eriskay.liston@sickkids.ca)
Laura Zahavich (laura.zahavich@sickkids.ca)
Dimitri Stavropoulos (james.stavropulos@sickkids.ca)
Rebekah Jobling (rebekah.jobling@sickkids.ca)
Raymond Kim (raymond.kim@sickkids.ca)
Miriam Reuter (miriam.reuter@sickkids.ca)
Anastasia Miron (anastasia.miron@sickkids.ca)
Erwin Oechslin (erwin.oechslin@uhn.ca)
Tapas Mondal (mondalt@mcmaster.ca)
Lynn Bergin (lynn.bergin@lhsc.on.ca)
John Smythe (john.smythe@kingstonhsc.ca)
Luis Altamirano-Diaz (luis.altamiranodiaz@lhsc.on.ca)
Jane Lougheed (jlougheed@cheo.on.ca)
Roderick Yao (roderick.yao@sickkids.ca)
Oyediran Akinrinade (oyediran.akinrinade@sickkids.ca)
Jeroen Breckpot (jeroen.breckpot@sickkids.ca)
Seema Mital (seema.mital@sickkids.ca)

Version: 3 Date: 08 Nov 2019

Author’s response to reviews:
RESPONSE TO EDITOR COMMENTS

Thank you for the final requests. Please see responses below.

1. Role of funding bodies?

Response: As requested, we have added the following statement to the Funding section, page 18, line 377: "The funding bodies played no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript."

2. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions.

RESPONSE: A clean version of the manuscript has been uploaded (along with tables, figures and supplementary materials that are unchanged).